

# GOPEN ACCESS

**Citation:** Johnson R, Burch KS, Hou K, Paciuc M, Pasaniuc B, Sankararaman S (2021) Estimation of regional polygenicity from GWAS provides insights into the genetic architecture of complex traits. PLoS Comput Biol 17(10): e1009483. https://doi. org/10.1371/journal.pcbi.1009483

**Editor:** Heather E. Wheeler, Loyola University Chicago, UNITED STATES

Received: April 24, 2021

Accepted: September 27, 2021

Published: October 21, 2021

**Peer Review History:** PLOS recognizes the benefits of transparency in the peer review process; therefore, we enable the publication of all of the content of peer review and author responses alongside final, published articles. The editorial history of this article is available here: https://doi.org/10.1371/journal.pcbi.1009483

**Copyright:** © 2021 Johnson et al. This is an open access article distributed under the terms of the Creative Commons Attribution License, which permits unrestricted use, distribution, and reproduction in any medium, provided the original author and source are credited.

**Data Availability Statement:** The BEAVR software is available as open source software at github.com/ bogdanlab/BEAVR.

**RESEARCH ARTICLE** 

# Estimation of regional polygenicity from GWAS provides insights into the genetic architecture of complex traits

Ruth Johnson<sup>1</sup>\*, Kathryn S. Burch<sup>2</sup>, Kangcheng Hou<sup>2</sup>, Mario Paciuc<sup>3</sup>, Bogdan Pasaniuc<sup>2,4,5,6°</sup>, Sriram Sankararaman<sup>1,2,5,6°</sup>\*

 Department of Computer Science, University of California, Los Angeles, California, United States of America, 2 Bioinformatics Interdepartmental Program, University of California, Los Angeles, California, United States of America, 3 Department of Statistics, Rice University, Houston, Texas, United States of America,
 Department of Pathology and Laboratory Medicine, David Geffen School of Medicine, University of California, Los Angeles, California, United States of America, 5 Department of Human Genetics, University of California, Los Angeles, California, United States of America, 6 Department of Computational Medicine, David Geffen School of Medicine, University of California, Los Angeles, California, United States of America

© These authors contributed equally to this work.

\* ruthjohnson@ucla.edu (RJ); sriram@cs.ucla.edu (SS)

# Abstract

The number of variants that have a non-zero effect on a trait (*i.e.* polygenicity) is a fundamental parameter in the study of the genetic architecture of a complex trait. Although many previous studies have investigated polygenicity at a genome-wide scale, a detailed understanding of how polygenicity varies across genomic regions is currently lacking. In this work, we propose an accurate and scalable statistical framework to estimate regional polygenicity for a complex trait. We show that our approach yields approximately unbiased estimates of regional polygenicity in simulations across a wide-range of various genetic architectures. We then partition the polygenicity of anthropometric and blood pressure traits across 6-Mb genomic regions (N = 290K, UK Biobank) and observe that all analyzed traits are highly polygenic: over one-third of regions harbor at least one causal variant for each of the traits analyzed. Additionally, we observe wide variation in regional polygenicity: on average across all traits, 48.9% of regions contain at least 5 causal SNPs, 5.44% of regions contain at least 50 causal SNPs. Finally, we find that heritability is proportional to polygenicity at the regional level, which is consistent with the hypothesis that heritability enrichments are largely driven by the variation in the number of causal SNPs.

# Author summary

The proportion of SNPs with nonzero effects on a trait, or polygenicity, is a key quantity used to describe the genetic architecture of a complex trait. Furthermore, identifying the specific genomic regions that contribute to trait variation requires an understanding of how the number of causal SNPs varies across regions of the genome (regional

**Funding:** RJ is partially supported by NSF grant DGE-1829071. SS is supported in part by NIH grant R35GM125055, NSF grant III-1705121, an Alfred P. Sloan Research Fellowship grant FG-2017-9480, and a gift from the Okawa Foundation. BP is supported in part by NIH grants R01HG009120, R01HG006399, and U01CA194393. The funders had no role in study design, data collection and analysis, decision to publish, or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

polygenicity). In this work, we propose a statistical framework to estimate regional polygenicity for a complex trait using marginal effect sizes from GWAS and LD information. We demonstrate in simulation and empirical data that our approach accurately and efficiently estimates regional polygenicity. We find that SNP-heritability is proportional to polygenicity both on the genome-wide and regional scale, suggesting that the observed differences in heritability across traits stem from differences in the underlying number of causal SNPs.

# Introduction

Polygenicity, i.e., the proportion of SNPs with nonzero effects on a trait, is a key quantity in efforts to understand the genetic architecture of complex traits. Accurate estimates of genome-wide polygenicity can be used to improve the prediction accuracy of polygenic risk scores [1, 2], quantify the strength of selection acting on a trait [3, 4], or better understand the biological complexity of the pathways driving disease risk [5, 6]. A major challenge in estimating polygenicity from genome-wide association study (GWAS) data arises due to the correlations between nearby SNPs, *i.e.* linkage disequilibrium (LD). In the presence of LD, methods for estimating polygenicity need to search over all possible causal status configurations at each SNP which, in turn, leads to an intractable computation for regions that harbor even a modest number of SNPs. Several methods implicitly model polygenicity in the context of phenotype prediction [7-11] whereas other methods explicitly aim to estimate polygenicity [2, 3], with recent methods overcoming the computational bottleneck by making simplifying model assumptions about the relationship between LD and polygenicity [2]. While all previous studies have focused on genome-wide polygenicity and its variation across traits [2], identification of genomic regions that are important for trait variation requires an understanding of how the number of causal SNPs varies across the genome (regional polygenicity).

In this work, we propose a statistical framework, Bayesian Estimation of Variants in a Region (BEAVR), to estimate regional polygenicity for a complex trait. Our approach estimates the proportion of causal variants in a given region ( $p_r$ ) using marginal effect sizes from GWAS and in-sample LD information. In this work, we define 'causal variants' as a set of variants measured in a given GWAS study that have either a nonzero effect on the trait or tag unmeasured variants through LD that also have a nonzero effect. This particular definition does not imply a causal biological relationship nor formal causation as defined in causal inference. Thus, the estimates of polygenicity are defined with respect to the set of variants in the analyzed GWAS. This is similar to the definition of SNP-heritability estimates which are also specific to each set of variants and cannot be extrapolated to other sets of SNPs [12–15].

The Bayesian model in BEAVR imposes a prior on the true SNP effect sizes where the probability of a non-zero true effect size at each SNP in the region is given by  $p_r$  [16, 17]. The observed GWAS effect sizes are obtained as a noisy combination of the unobserved true SNP effect sizes [18, 19]. We use Markov chain Monte Carlo (MCMC) [20] to approximate the posterior probability of the regional polygenicity parameter. This inference problem is computationally challenging as it requires disentangling correlations between SNPs due to LD. Leveraging the insight that the genetic architectures of most traits are likely to be sparse (so that most SNPs are not causal), we obtain a substantially more efficient MCMC algorithm that allows us to infer regional polygenicity across a large number of SNPs.

We validate our approach using extensive simulations and find that our method accurately estimates polygenicity in realistic settings; BEAVR estimates yield a relative bias < 2% across all simulations whereas existing methods obtain biased estimates particularly in simulations with high degrees of polygenicity (*i.e.*  $p_r > 5\%$ ). Next, we estimate regional polygenicity across 6-Mb regions for five quantitative anthropometric and blood pressure traits in the UK Biobank (N = 290, 641 unrelated British individuals) restricting to genotyped SNPs with MAF > 1%. Consistent with previous works [2], we find that all analyzed traits are highly polygenic at the genome-wide scale: over one-third of regions harbor at least one causal SNP across all traits. The proportion of regions containing at least one causal SNP (typically defined as regions with significant heritability) has been used as a proxy for polygenicity in earlier studies [12, 21]; we find that the proportion of regions containing at least one causal SNP is much higher than the estimated polygenicity. For example, while 79.6% of regions contain at least one causal SNP for height, the genome-wide polygenicity is estimated to be 3.07%. Additionally, we observe wide variation in regional polygenicity: on average across all analyzed traits, 48.9% of regions contain at least 5 causal SNPs while 5.44% of regions contain at least 50 causal SNPs, demonstrating the additional information provided from estimates of regional polygenicity. Finally, we find that within traits, regional SNP-heritability is proportional to regional polygenicity, suggesting that variation in heritability across the genome is largely driven by variation in the number of causal SNPs.

# Materials and methods

#### Generative model

We assume that the trait measured in individual *i*,  $y_i$ , is a linear function of standardized genotypes  $\mathbf{x}_i = (x_{i,1}, \dots, x_{i,M})$  measured at *M* SNPs with true SNP effect sizes  $\boldsymbol{\beta} = (\beta_1, \dots, \beta_M)$  and an independent additive noise term  $\epsilon_i$ .

$$\mathbf{y}_i = \sum_{m=1}^M \beta_m \mathbf{x}_{i,m} + \epsilon_i, i \in \{1, \cdots N\}$$
(1)

$$\epsilon_i \stackrel{iid}{\sim} \mathcal{N}(0, \sigma_e^2) \tag{2}$$

We model a non-infinitesimal trait architecture in which a subset of the *M* SNPs are causal by imposing a spike-and-slab prior on the causal effect sizes  $\beta$  [2, 3, 7]. We represent the causal statuses across the SNPs as  $c = (c_1, \dots, c_M)$ . Here,  $c_m = 1$  if SNP *m* is a causal SNP with probability *p* and 0 otherwise. Thus, *p* denotes the proportion of causal SNPs or the *polygenicity*.

The Gaussian slab is parametrized with mean 0 and variance  $\frac{h_{GW}^2}{Mp}$  where  $h_{GW}^2$  is the genomewide heritability. We draw independent Gaussian random variables for each of the *M* SNPs:  $\gamma = (\gamma_1, \dots, \gamma_M)$ . The effect size  $\beta_m$  is  $\gamma_m$  if SNP *m* is causal and 0 otherwise.

$$\gamma_m \sim \mathcal{N}(0, \frac{h_{GW}^2}{Mp}) \tag{3}$$

$$\beta_m \mid c_m, \gamma_m = \begin{cases} \gamma_m & \text{if } c_m = 1\\ 0 & \text{if } c_m = 0 \end{cases}$$
(4)

We model the conditional distribution of the GWAS effect sizes given the true effect sizes where  $\hat{\beta}_m$  is the estimated marginal effect size of SNP *m* for the trait.

$$\hat{\boldsymbol{\beta}}|\boldsymbol{\beta} \sim \mathcal{N}(\boldsymbol{V}\boldsymbol{\beta}, \boldsymbol{V}\sigma_e^2)$$
 (5)

Here the covariance matrix is parametrized by the environmental noise  $\sigma_e^2$  and the correlations among SNPs, *i.e.* the linkage disequilibrium (LD) matrix *V*. The variance of the environmental noise term is parameterized by  $\sigma_e^2 = \frac{1-h_{GW}^2}{N}$ , where *N* is the number of individuals in the study.

We impose a symmetric Beta prior on the polygenicity parameter, *p*.

$$p \sim Beta(\alpha, \alpha)$$
 (6)

In practice, we use  $\alpha = 0.2$  to put a higher weight on the tails of the Beta distribution. The full graphical model can be found in <u>S1 Fig</u>.

In this work, we focus on accurately estimating the proportion of causal variants in a given region r (regional polygenicity,  $p_r$ ). We assume that the above proposed genome-wide generative model holds when applied only within a specific region of the genome. This includes modeling the heritability only within that region  $(h_r^2)$  instead of the genome-wide heritability  $(h_{GW}^2)$ . Modeling each region separately also assumes that there are no correlations between regions, such as correlations due to long-range LD. This assumption is reasonable when regions are chosen to correspond to LD blocks or when regions are sufficiently large such that correlations with adjacent regions may be ignored. Therefore, the LD matrix used in the regional model would only be the LD computed from SNPs within that particular region  $(V_r)$ . Additionally, although our framework naturally estimates the SNP effect sizes and posterior inclusion probabilities (*i.e.*, the probability that a given SNP is causal), we focus in this work on the posterior probability of  $p_r$ .

The posterior probability of the model parameters of interest  $(p_r, \gamma_r, c_r)$  for a given region r is given by:

$$P(p_r, \boldsymbol{\gamma}_r, \boldsymbol{c}_r \mid \boldsymbol{\beta}_r, \alpha, h_r^2) \propto P(p_r \mid \alpha) P(\boldsymbol{c}_r \mid p_r) P(\boldsymbol{\gamma}_r \mid h_r^2, p_r) P(\boldsymbol{\beta}_r \mid \boldsymbol{\gamma}_r, \boldsymbol{c}_r, h_r^2)$$
(7)

#### Inference

We use Markov Chain Monte Carlo (MCMC) to approximate the posterior probability as defined in Eq.7. Specifically, we derive a Gibbs sampler [22] to sample from the posterior distribution of the regional polygenicity  $p_r$  and latent variables ( $c_r$ ,  $\gamma_r$ ). The method takes as input the marginal effect sizes from GWAS for a single trait in a region  $r(\hat{\beta}_r)$ , the matrix of SNP correlations or LD per region ( $V_r$ ), an estimate of the SNP heritability in that region ( $h_r^2$ ), and the sample size of the GWAS (N). As output, we estimate the posterior probability of the regional polygenicity for region  $r(p_r)$ .

**Transforming GWAS effect sizes.** To facilitate efficient inference, we transform the marginal effects from GWAS:  $\tilde{\boldsymbol{\beta}}_r \equiv V_r^{-\frac{1}{2}} \boldsymbol{\beta}_r$ . The conditional probability of these transformed effects is given by:

$$\tilde{\boldsymbol{\beta}}_{r} \mid \boldsymbol{\beta}_{r} \sim \mathcal{N}\left(\boldsymbol{V}_{r}^{\frac{1}{2}} \boldsymbol{\beta}_{r}, \boldsymbol{I}_{M_{r}} \sigma_{e}^{2}\right)$$

Here,  $I_{M_r}$  is the identity matrix of size  $M_r \times M_r$  where  $M_r$  is the number of SNPs in region r. We note that this is a one-time transformation that is performed before running the sampler. These transformed effects can be efficiently computed and stored for each genomic region. **Sampling**  $\gamma_r$  and  $c_r$ . We recall that the true effect size at SNP *m* in region *r* ( $\beta_{r,m}$ ) is given by a spike-and-slab prior parametrized by the causal effect size ( $\gamma_{r,m}$ ) and the causal status at that SNP ( $c_{r,m}$ ) (see Eq 4). We choose to sample  $\gamma_{r,m}$  and  $c_{r,m}$  together in a block step in the Gibbs sampler update.

Let  $\theta_r = \{(\gamma_{\neg r,m}, \boldsymbol{c}_{\neg r,m}), h_r^2, p_r, \alpha\}$ , where  $\gamma_{\neg r,m}$  denotes all effect sizes except for the effect of the  $m^{th}$  SNP; this similarly follows for  $\boldsymbol{c}_{\neg r,m}$ .

$$P(\gamma_{r,m}, c_{r,m} \mid \boldsymbol{\theta}_r, \boldsymbol{\beta}_r) = P(\gamma_{r,m} \mid c_{r,m}, \boldsymbol{\theta}_r, \boldsymbol{\beta}_r) P(c_{r,m} \mid \boldsymbol{\theta}_r, \boldsymbol{\beta}_r)$$

We are interested in the posterior probability of the causal effect size  $\gamma_{r,m}$  when  $c_{r,m} = 1$  since  $P(\gamma_{r,m} \mid c_{r,m} = 0) = 0$  due to the spike-and-slab prior. This can be expressed as:

$$P(\gamma_{r,m} \mid c_{r,m} = 1, \boldsymbol{\theta}_r, \boldsymbol{\beta}_r) \propto P(\boldsymbol{\beta}_r \mid \gamma_{r,m}, c_{r,m} = 1, \boldsymbol{\theta}_r) P(\gamma_{r,m} \mid c_{r,m} = 1, \boldsymbol{\theta}_r)$$

Working with the transformed GWAS effect sizes, the posterior distribution of  $\gamma_{r,m}$  becomes univariate Gaussian with the following mean and variance. Here we denote  $\mathbf{r}_{r,m} = \tilde{\boldsymbol{\beta}}_r - \mathbf{V}_r^{\frac{1}{2}} \gamma_r \circ \mathbf{c}_r + \mathbf{V}_{r,m}^{\frac{1}{2}} \gamma_{r,m} \mathbf{c}_{r,m}$ , which is the residual from subtracting the effects of all SNPs except for SNP *m* (here  $\mathbf{V}_{r,m}^{\frac{1}{2}}$  denotes column *m* of the matrix  $\mathbf{V}_r^{\frac{1}{2}}$ ). We define  $\sigma_{r,g}^2 = \frac{h_r^2}{M_r p_r}$  and  $\sigma_s^2 = \frac{1-h_r^2}{M_r}$  for the region-specific model. See S1 File for full derivation details.

$$P(\gamma_{r,m} \mid c_{r,m} = 1, \boldsymbol{\theta}_r, \tilde{\boldsymbol{\beta}}_r) = \mathcal{N}(\gamma_{r,m}; \mu_{r,m}, \sigma_{r,m}^2)$$

$$\frac{1}{\sigma_{r,m}^2} = \frac{1}{\sigma_{r,g}^2} + \frac{1}{\sigma_e^2} \boldsymbol{V}_{r,m}^{\frac{1}{2}} \boldsymbol{V}_{r,m}^{\frac{1}{2}}$$

$$\mu_{r,m} = \sigma_{r,m}^2 \frac{1}{\sigma_e^2} \boldsymbol{r}_{r,m}^{\top} \boldsymbol{V}_{r,m}^{\frac{1}{2}}$$
(8)

We sample  $c_{r,m} \mid \boldsymbol{\theta}_r, \tilde{\boldsymbol{\beta}}_r$  from a Bernoulli distribution with parameter  $P(c_{r,m} = 1 \mid \boldsymbol{\theta}_r, \tilde{\boldsymbol{\beta}}_r)$ :

$$\begin{split} P(c_{r,m} &= 1 \mid \boldsymbol{\theta}_r, \tilde{\boldsymbol{\beta}}_r) &= \int P(c_{r,m} = 1, \gamma_{r,m} \mid \boldsymbol{\theta}_r, \tilde{\boldsymbol{\beta}}_r) d\gamma_{r,m} \\ &= \int \frac{P(\tilde{\boldsymbol{\beta}}_r \mid \gamma_{r,m}, c_{r,m} = 1, \boldsymbol{\theta}_r) P(\gamma_{r,m}, c_{r,m} = 1 \mid \boldsymbol{\theta}_r)}{P(\tilde{\boldsymbol{\beta}}_r \mid \boldsymbol{\theta}_r)} d\gamma_{r,m} \\ &= \frac{(p_r) \sqrt{\frac{\sigma_{r,m}^2}{\sigma_{r,g}^2}} \exp\left\{\frac{1}{2\sigma_{r,m}^2} \mu_{r,m}^2\right\}}{(p_r) \sqrt{\frac{\sigma_{r,m}^2}{\sigma_{r,g}^2}} \exp\left\{\frac{1}{2\sigma_{r,m}^2} \mu_{r,m}^2\right\} + (1 - p_r)} \\ &= d_{r,m} \end{split}$$

**Sampling**  $p_r$ . The complete conditional posterior distribution of  $p_r$  depends not only on the causal status of each SNP ( $c_{r,m}$ ), but also on the latent variable ( $\gamma_{r,m}$ ) since  $p_r$  parametrizes the variance term of  $\gamma_{r,m}$ . We sample from this distribution using a random-walk Metropolis-Hastings step [20]. We use a Beta distribution as a proposal distribution:

$$p_r^* \sim Q(p_r^* | p_r)$$
  
=  $Beta(\alpha + Cp_r, \alpha + C(1 - p_r))$ 

Here, *C* is a constant that controls the variance of the proposal distribution. In practice we found that C = 10 yield effective mixing.

# Leveraging sparsity of the genetic architecture to improve the computational efficiency

The key computational bottleneck in the Gibbs sampling scheme involves computing the mean of the posterior distribution of the causal effect size at SNP  $m(\mu_{r,m} \text{ in Eq 8})$ . Specifically, the matrix computations associated with the residual term,  $\mathbf{r}_{r,m} = \tilde{\boldsymbol{\beta}}_r - V_r^{\frac{1}{2}} \gamma_r \circ \mathbf{c}_r + V_{r,m}^{\frac{1}{2}} \gamma_r \sigma_r, \mathbf{c}_r, \mathbf{r}_r, \mathbf{r}_r,$ 

Writing out the posterior mean term and expanding, we have:

$$\begin{split} \mu_{r,m} &= \frac{\sigma_{r,m}^2}{\sigma_e^2} \boldsymbol{r}_{r,m}^\top \boldsymbol{V}_{r,m}^{\frac{1}{2}} \\ &= \frac{\sigma_{r,m}^2}{\sigma_e^2} \left[ \tilde{\boldsymbol{\beta}}_r - \boldsymbol{V}_r^{\frac{1}{2}} \gamma_r \circ \boldsymbol{c}_r + \boldsymbol{V}_{r,m}^{\frac{1}{2}} \gamma_{r,m} \boldsymbol{c}_{r,m} \right]^\top \boldsymbol{V}_{r,m}^{\frac{1}{2}} \\ &= \frac{\sigma_{r,m}^2}{\sigma_e^2} \left[ \tilde{\boldsymbol{\beta}}_r - \sum_{m \neq l}^{M_r} \boldsymbol{V}_{r,l}^{\frac{1}{2}} \boldsymbol{\gamma}_{r,l} \boldsymbol{c}_{r,l} \right]^\top \boldsymbol{V}_{r,m}^{\frac{1}{2}} \\ &= \frac{\sigma_{r,m}^2}{\sigma_e^2} \left[ \tilde{\boldsymbol{\beta}}_r^\top \boldsymbol{V}_{r,m}^{\frac{1}{2}} - \sum_{l \neq m, c_{r,l}=1}^{M_r} \boldsymbol{V}_{r,l}^{\frac{1}{2}} \boldsymbol{V}_{r,m}^{\frac{1}{2}} \boldsymbol{V}_{r,m}^{\frac{1}{2}} \right] \end{split}$$

The first term,  $\tilde{\boldsymbol{\beta}}_r^\top \boldsymbol{V}_{r,m}^{\frac{1}{2}}$ , is composed of the vector of GWAS effect sizes and a vector of the LD matrix corresponding to the  $m^{th}$  SNP, neither of which are updated within the sampler. Second, the term  $\boldsymbol{V}_{r,l}^{\frac{1}{2}} \boldsymbol{V}_{r,m}^{\frac{1}{2}}$  can also be pre-computed since it is only the product of two columns within the LD matrix. Aside from the variance terms at the beginning of the equation, which are only scalars, the only term that varies at each iteration of the sampler is  $\gamma_{r,l} c_{r,l}$  since both the effect size and causal status need to be re-sampled at each iteration. Since this term is wrapped in a summation over  $M_r$  SNPs, the complexity of computing  $\mu_{r,m}$  is currently  $\mathcal{O}(M_r)$ . However, even with this simplification, the overall complexity of the sampler is  $\mathcal{O}(M_r^2)$  since this mean term must be computed at every SNP at every iteration.

To further simplify the computation, we can leverage the observation that most complex traits contain only a small proportion of causal SNPs  $(K_r)$  in each region. As the sampler converges to the stationary distribution, we would expect the causal status vector  $(\mathbf{c}_r)$  to be sparse, where  $K_r \ll M_r$ . When this occurs, the summation term will only include a few non-zero terms. By only subtracting the non-zero terms, this term is simply reduced to the number of causal variants and the complexity becomes  $\mathcal{O}(K_r)$ . Even though this computation must be

done at each SNP, the overall complexity of the sampler is only  $\mathcal{O}(K_rM_r)$  which is tractable under the assumption of  $K_r \ll M_r$ .

#### Simulation analysis

Simulations for marginal effects using LD information. Using pre-computed LD information, we generated marginal effect sizes for a given region from synthetic GWAS that reflect a variety of genetic architectures. We denote the number of SNPs in a region as  $M_r$  and the regional polygenicity as  $p_r$ . We denote the causal indicator status of each SNP in each region as  $c_{r,m} \in \{0, 1\}$ , where  $c_{r,m} = 1$  if the  $m^{th}$  SNP is causal and 0 otherwise for  $m = 1, \dots, M_r$  and regions  $r = 1, \dots, R$ .

The causal status of a SNP is generated from:

$$c_{r,m} \sim Ber(p_r)$$

If  $c_{r,m} = 1$ , the effect size of SNP *m* within the  $r^{th}$  region is drawn from a univariate Gaussian distribution with mean 0 and variance equal to the regional heritability  $(h_r^2)$  divided by the number of casual SNPs:

$$eta_{\mathrm{r,m}} \sim \left\{ egin{array}{ll} 0, & c_{\mathrm{r,m}} = 0, \\ & & \\ \mathcal{N}\Big(0, rac{h_{\mathrm{r}}^2}{M_{\mathrm{r}} p_{\mathrm{r}}}\Big), & c_{\mathrm{r,m}} = 1 \end{array} 
ight.$$

Marginal association statistics for the region are then generated from the following model:

$$\hat{\boldsymbol{\beta}}_r \mid \boldsymbol{\beta}_r \sim \mathcal{N}(\boldsymbol{V}_r \boldsymbol{\beta}_r, \boldsymbol{V}_r \sigma_e^2)$$

Here, the environmental noise is a function of the sample size and heritability of the trait,  $\sigma_e^2 = \frac{1-h_r^2}{N}$ . We use regional LD computed with genotypes from 337, 205 unrelated (less related than third-degree relatives), white, British individuals ( $M_r = 1$ , 000 array SNPs) from the UK Biobank [23]. The LD matrix for a region is computed as  $V_r = \frac{X_r^T X_r}{N}$ , where  $X_r$  is the genotype matrix using only SNPs within region r.

Using the framework above, we generated marginal effect sizes where we varied the regional polygenicity from  $p_r = 0.005$ , 0.01, 0.05, and 0.10, genome-wide heritability from  $h_{GW}^2 = 0.10$ , 0.25, and 0.50, and the sample size from N = 50K, 500K, 1M individuals, which is comparable to the sample sizes of many current GWAS studies [24, 25]. For each simulated region, we set the number of SNPs per region to 1, 000. For the regional heritability parameter, we used the simulated genome-wide heritability scaled by the number of SNPs in the region,

 $M_r$ , and the number of SNPs on the array, M:  $h_r^2 = \frac{h_{GW}^2 M_r}{M}$ .

To estimate the regional polygenicity, we ran BEAVR for 1,000 iterations with a burn-in of 250 iterations. We used the same LD information that was used for simulation (*i.e.* "in-sample" LD). We also computed regional polygenicity using GENESIS [2]. We ran GENESIS using the default parameter settings and LD information from 1000 Genomes [26]. We used both the 2-component and 3-component settings when running GENESIS. We note that the implementation of GENESIS uses the 1000 Genomes LD matrix as a default and there is no option to specify an alternative LD matrix. We averaged the performance of each method across 100 replicates.

Simulations for marginal effects computed from individual genotype and phenotype data. Using SNP data (M = 9, 564 array SNPs from chromosome 22, N = 337K individuals) from a group of unrelated, self-identified British, white ancestry individuals from the UK

Biobank [23], we simulated marginal effects by generating phenotypes from real genotype array data. For this analysis, the set of unrelated individuals is defined as pairs of individuals with kinship coefficient  $<\frac{1}{2}^{(9/2)}$  (greater than third-degree relatives) [23]. Then we performed ordinary least squares to estimate the marginal effect size of each SNP. Given the standardized genotype matrix X and the genome-wide SNP heritability  $h_{GW}^2$ , phenotypes are generated as follows.

We set the genome-wide proportion of causal variants to be p = 0.01. We denote the causal indicator status of each SNP as  $c_m \in \{0, 1\}$ , where  $c_m = 1$  if the  $m^{th}$  SNP is causal and 0 otherwise for  $m = 1, \dots, M$ . Standardized effects and phenotypes are generated from the following model. Note that  $\sigma_m^2 = 0$  if  $c_m = 0$ .

$$\sigma_m^2 = c_m \frac{h_{GW}^2}{Mp}$$
$$(\beta_1, \cdots, \beta_M)^\top \sim \mathcal{N}(0, \operatorname{diag}(\sigma_1^2, \cdots, \sigma_M^2))$$
$$(y_1, \cdots, y_N)^\top \mid \boldsymbol{\beta} \sim \mathcal{N}(\boldsymbol{X}\boldsymbol{\beta}, (1 - h_{GW}^2)\boldsymbol{I}_N)$$

Finally, given the phenotypes for all individuals,  $\boldsymbol{y} = (\boldsymbol{y}_1, \dots, \boldsymbol{y}_N)^{\top}$  and genotypes  $\boldsymbol{X} = (\boldsymbol{x}_1^{\top}, \dots, \boldsymbol{x}_N^{\top})^{\top}$ , we compute marginal association statistics through the OLS estimator,  $\hat{\boldsymbol{\beta}} = \frac{1}{N} \boldsymbol{X}^{\top} \boldsymbol{y}$ .

We generated 100 sets of marginal effect sizes where we fixed p = 0.01 and  $h_{GW}^2 = 0.50$ . We then estimated the regional polygenicity within each 6-Mb window for chromosome 22 (M = 9,564 array SNPs) using BEAVR. This windowing formed 6 consecutive regions. We used HESS (Heritability Estimator from Summary Statistics) [12], a method for estimating regional heritability at a single region from GWAS summary statistics, to estimate the regional heritability which is then used as input for BEAVR. HESS is run with all default parameters and the same LD matrices used in the simulation framework (*i.e.* in-sample LD). We finally ran BEAVR for 1,000 iterations with a burn-in of 250 iterations and using the same LD information that was used for simulation.

#### Analysis of UK Biobank phenotypes

We estimated the partitioned polygenicity for five complex traits in the UK Biobank [23] across 6-Mb windows. We limited our analyses to unrelated individuals with self-identified British, white ancestry. Here, the set of unrelated individuals is defined as pairs of individuals with kinship coefficient  $<\frac{1}{2}^{(9/2)}$  (greater than third-degree relatives) [23]. We additionally excluded individuals with putative sex chromosome aneuploidy. All genotypes were standard-ized, where for each SNP *m* and individual *n*, we computed  $x_{nm} = (g_{nm} - 2f_m)/\sqrt{2f_m(1 - f_m)}$ , where  $g_{nm} \in \{0, 1, 2\}$  is the number of minor alleles and  $f_m$  is the in-sample minor allele frequency (MAF). We then used PLINK [27] (https://www.cog-genomics.org/plink2) to exclude SNPs with MAF < 0.01, genotype missingness > 0.01, and SNPs that fail the Hardy-Weinberg test at significance threshold  $10^{-7}$ . We obtained a final set of N = 290, 641 individuals for our analyses.

Marginal association statistics were computed through OLS using PLINK. Age, sex, and the top 20 genetic PCs were used as covariates in the regression, where these top 20 PCs were precomputed by the UK Biobank from a superset of 488, 295 individuals. Additional covariates were used for waist-to-hip ratio (adjusted for body mass index (BMI)) and diastolic/systolic blood pressure (adjusted for cholesterol-lowering medication, blood pressure medication, insulin, hormone replacement therapy, and oral contraceptives). The genome is then divided into 6-Mb windows. Using HESS [12], we estimated the regional heritability within each window for each trait. HESS is run with all default parameters specified and in-sample LD. Using BEAVR and the computed regional heritability estimates, we estimated the regional polygenicity in each 6-Mb window. To initialize the MCMC sampler, we must set initial values for the vector of causal statuses, causal effect sizes, and regional polygenicity ( $c_r$ ,  $\gamma_r$ ,  $p_r$ ). For each SNP *m*, if the z-score estimated from GWAS is  $\geq$  3.5, then  $c_{r,m}$  is initialized to 1 and 0 otherwise. Each causal effect size is drawn from the prior distribution (see Eq 3). The initial value of  $p_r$  is set to the proportion of 1's in the initialized causal status vector. We ran the Gibbs sampler for 1,000 iterations and the first 250 samples were discarded as burn-in. For each region, we computed the posterior mean and posterior standard deviation for  $p_r$  from the MCMC samples.

#### Annotations in regression analysis

We performed a multivariate regression of the heritability on the estimated number of SNPs from BEAVR, the number of causal SNPs, and genomic annotations within a region. The genomic annotations include the number of genes, median *B* value (a measure of background selection), and functional annotations [28]. We computed the number of protein coding genes within a region using the protein coding gene sets that have been defined in previous work [29]. If a gene body overlapped two regions, we included the presence of the gene in both regions. Using previously computed *B* values [30], we computed the median *B* value of all the SNPs in a region. This quantity was used as the annotation value for that particular region. We additionally included a combination of binary and continuous functional annotations [28]. For each region, we computed the median annotation value for continuous annotations and the proportion of variants with a binary annotation.

# Results

#### Simulations

We compare BEAVR to GENESIS [2], an approach that employs a spike-and-slab mixture model to capture both large and small effect sizes at causal SNPs in order to estimate polygenicity at a genome-wide scale (see <u>Materials and methods</u>). To be applicable in genome-wide settings, GENESIS assumes that LD patterns are independent of the probability of a SNP belonging to different mixture components which, in turn, leads to a scalable algorithm. As shown in Fig 1, BEAVR obtains approximately unbiased estimates of polygenicity across each scenario (relative bias < 2% across the simulations). Both the two and three mixture component models from GENESIS obtain relatively unbiased estimates when the true polygenicity is low but demonstrates a severe downward bias in the high polygenicity setting (relative bias > 70% when  $p_r = 0.10$ ). This observation is consistent with our hypothesis that not fully modeling LD limits the ability of GENESIS to accurately estimate parameters, consistent with previously reported downward bias when GENESIS was run with external LD information [2].

Next, we assessed the robustness of our approach to sample size and heritability. We vary the genome-wide heritability to be 0.10 and 0.25 and the sample size to be 50K and 1 million individuals (Fig 2) to fully explore the limitations of our method. We note that when the regional polygenicity  $p_r$  is high, BEAVR demonstrates a downward bias either when sample sizes are relatively small (N = 50K individuals) (relative bias 56% and 80% for  $p_r = 0.05$  and  $p_r = 0.10$  when  $h_{GW}^2 = 0.50$ ) or when the heritability is low ( $h_{GW}^2 = 0.10$ ) (relative bias 54% and 73% for  $p_r = 0.05$  and  $p_r = 0.10$  for N = 500K). These biases likely arise due to the causal effect sizes being similar in magnitude to the environmental noise, making it difficult to



**Fig 1. BEAVR is relatively unbiased in simulated data.** We ran 100 replicates (M = 1, 000 SNPs, N = 500K individuals) where the genome-wide heritability was set to  $h_{GW}^2 = 0.5$  and the true polygenicity of the region was  $p_r = 0.005, 0.01, 0.05, 0.10$ . We compared BEAVR to GENESIS-M2 and GENESIS-M3 which employs a spike-and-slab model with either 2 or 3 components (point-mass and either 1 or 2 slabs). All methods are unbiased when the polygenicity is low ( $p_r = 0.005, 0.01$ ). However, when polygenicity is higher ( $p_r = 0.05, 0.10$ ), both GENESIS-M2 and GENESIS-M3 are severely downward biased whereas BEAVR provides unbiased estimates across all settings. Dashed red lines denote true regional polygenicity values in each setting.

correctly identify the causal status of a SNP. Thus, we recommend applying BEAVR to heritable traits measured in large sample sizes.

Next, we performed simulations where GWAS marginal effects are computed from phenotypes simulated from individual genotypes and the regional heritability is estimated directly from the data. Specifically, we simulate phenotypes using individual genotypes for N = 337K individuals from the UK Biobank. Each phenotype is simulated to have  $h_{GW}^2 = 0.50$  and polygenicity p = 0.01. We limit our simulations to SNPs from chromosome 22 (M = 9, 564 SNPs) as each chromosome would be analyzed separately in real data analyses. We then estimate the marginal effect sizes. We divide the simulated data into consecutive regions of 6-Mb for a total of 6 regions, where each region contains 1,000 SNPs on average. We use estimates of regional heritability from GWAS marginal effects (using HESS [12]; see Materials and methods) as input to BEAVR. We find that BEAVR obtains relatively unbiased estimates of polygenicity across all regions (Fig 3A; relative bias < 2% across simulations). These simulations indicate that the polygenicity estimates obtained by BEAVR are robust to heritability estimates that are used as input as well as when LD spans regions. The LD does not significantly affect the estimates likely because the correlation due to LD tends to diminish with genomic distance.

We also explored the robustness of BEAVR to the number of SNPs in the region. Using a simulated GWAS with genome-wide heritability  $h_{GW}^2 = 0.50$ , sample size N = 500K, and polygenicity  $p_r = 0.01$ , we vary the size of the region from  $M_r = 500$ , 1K, 5K SNPs. From Fig 3B, we can see that the estimates of  $p_r$  tend to be unbiased across regions of various sizes although the standard errors tend to increase in smaller regions (relative bias 13%, 1%, and



**Fig 2. BEAVR is relatively unbiased across various genetic architectures.** We ran 100 replicates where we vary the genome-wide heritability to be  $h_{GW}^2 = 0.10, 0.25$ , 0.5, the polygenicity of the region to be  $p_r = 0.005, 0.01, 0.05, 0.10$ , and the sample size N = 50K, 500K, 1 million individuals. We compared BEAVR to GENESIS-M2 (2-component) and GENESIS-M3 (3-component). The x-axis denotes the simulated values for the regional polygenicity and the y-axis denotes the estimated values across 100 replicates. Dashed red lines denote the true regional polygenicity value in each setting.

1.2% for  $M_r = 500$ , 100, 5000 SNPs). This trend occurs because larger regions have a higher number of SNPs to inform the posterior distribution, meaning that there will be higher certainty in the posterior estimates. Additionally, if a region is small, there is a larger impact on the estimated polygenicity when misidentifying causal SNPs due to the small denominator of SNPs in the region. For example, misidentifying a single causal SNP from a set of 10 SNPs will have a greater impact on the bias of polygenicity estimates compared to a set of 1,000 SNPs. These results suggest that BEAVR could potentially be applied to regions of varying length and be used to estimate regional polygenicity around genes or within larger LD blocks.

We next assess the sensitivity of our results when using different hyper-parameters for our prior on the polygenicity parameter  $p_r$ . Using a simulated GWAS with genome-wide heritability  $h_{GW}^2 = 0.50$ , sample size N = 500K, and polygenicity  $p_r = 0.01$ , we vary our choice of hyper-parameter for the prior on  $p_r$ :  $\alpha = 0.2$ , 1, 2. We find that the accuracy of our results is relatively robust to the choice of prior (Fig 3C); we use  $\alpha = 0.2$  for all subsequent analyses.



Fig 3. BEAVR is robust in realistic settings. (A) Using SNP data from chromosome 22 (M = 9, 564 array SNPs, N = 337K individuals), we simulated 100 replicates where the genome-wide heritability was  $h_{GW}^2 = 0.50$  and p = 0.01. We divided the data into 6-Mb consecutive regions for a total of 6 regions and estimated the regional heritability using external software (HESS [12]). Using BEAVR and the estimated regional heritability, we estimated the regional polygenicity to be unbiased across all regions. (B) We ran 100 replicates where the genome-wide heritability is fixed  $h_{GW}^2 = 0.50$ , polygenicity  $p_r = 0.01$ , sample size N = 500K, and then varied the number of SNPs in the region from M = 500, 1K, 5K SNPs. We used BEAVR to estimate the polygenicity in each region and found our results to be unbiased across all regions. (C) We set the genome-wide heritability to  $h_{GW}^2 = 0.50$ , regional polygenicity  $p_r = 0.01$ , and sample size N = 500K. We find that the accuracy of our results is invariant to our choice of prior hyper-parameter ( $\alpha$ ).

Although we assume that causal effect size distributions follow a Gaussian distribution, there are likely traits which do not follow this assumption. We next evaluate the performance of BEAVR when the true causal effect sizes deviate from the assumption of normality. In S2 Fig, we show the distribution of our estimates of regional polygenicity across 3 causal effect size distributions defined by mixture of Gaussian distributions with the following set of variance components and mean 0:  $[1 \times 10^{-3}, 1 \times 10^{-4}]$ ;  $[1 \times 10^{-4}, 1 \times 10^{-5}]$ ;  $[1 \times 10^{-3}, 1 \times 10^{-4}, 1 \times 10^{-5}]$ . We assume a sample size of N = 500K, total regional polygenicity  $p_r = 0.01$ , and assume the true heritability of the region is known. The number of causal SNPs are spread equally amongst all the mixture components (*e.g.*  $1 \times 10^{-3}$ ), our estimates are relatively unbiased. However, for distributions with smaller variance components (*e.g.*  $1 \times 10^{-3}$ ), we start to see a downward bias proportional to the fraction of SNPs drawn from the distribution with the smaller variance component(s). Thus, it is not necessarily the exact shape of the distribution of effect sizes, but the magnitude of the causal effect sizes, which affects the accuracy of the estimates.

#### Effect of out-of-sample LD

Although we recommend using in-sample LD when computing estimates of regional polygenicity, we also investigate the scenario where only LD derived from a reference panel is available. We simulate two scenarios: i) reference panel LD is computed from genotypes from individuals of a similar continental population as the target GWAS population but from a separate study (*e.g.* European ancestry individuals from the 1000 Genomes Project); ii) reference panel LD is computed with genotypes from a specific cohort/study and the target GWAS is also conducted with a subset of data from the same the cohort/study or a different version of the study (*e.g.* 'White, British' individuals from the UK Biobank). This second scenario closely reflects situations where many groups separately apply for freezes of data from the same study yet share GWAS summary statistics across applications.

We simulate the first scenario by simulating 1,000 GWAS regions with M = 1,000 SNPs, regional polygenicity  $p_r = 0.01$ , regional heritability,  $h_r^2 = 0.0001$  (corresponding to a genomewide heritability of 0.50), a sample size of N = 500K, and use a LD matrix computed from 337, 205 genotypes from unrelated individuals within the 'White, British' population from the UK Biobank [23]. However, inference is then performed using a reference panel derived LD matrix computed from 503 European ancestry individuals from the 1000 Genomes Project [26]. We find that when using LD from these separate studies, BEAVR fails to accurately estimate the regional polygenicity (S3 Fig). Although the reference panel is constructed using individuals of European ancestry, these individuals were sampled from multiple subcontinental ancestries in Europe (*e.g.* Italy, Spain, Finland). In comparison, the target GWAS population from the UK Biobank is ancestrally homogeneous since it is limited to 'White, British' individuals within the UK.

The second scenario uses the same simulation parameters as above, except the GWAS effect sizes are computed using a LD matrix derived from 168,602 individuals from the unrelated, 'White, British' population within the UK Biobank. Inference is then performed using LD estimated from a separate, non-overlapping set of 168,602 individuals also from the unrelated, 'White, British' population within the UK Biobank. When using LD computed from a separate set of individuals from the same study, we find that our estimates are approximately unbiased (S3 Fig). Our findings show that one can perform inference using a reference panel constructed from a separate set of individuals than used in the GWAS when both sets of individuals are from the same study (*e.g.* UK Biobank). These findings suggest that LD reference panels cannot solely be matched based on the continental ancestry level but need to be matched on a much finer scale. Additionally, differences in study designs between the genotypes used for the LD reference panel and the genotypes used when performing the GWAS may also contribute to discrepancies between the estimated LD structure.

#### **Computational efficiency**

BEAVR uses Gibbs sampling [20] to estimate the posterior probability of the regional polygenicity parameter. A naive implementation of the Gibbs sampler has a per-iteration computational complexity of  $\mathcal{O}(M_r^2)$ , where  $M_r$  is the number of SNPs in the region. By leveraging the expected sparsity of the causal status at each SNP, we can improve the run-time of the algorithm to  $\mathcal{O}(M_rK_r)$ , where  $K_r$  is the number of causal SNPs in the region. Fig 4A shows that this improvement leads to a 12-fold improvement in run-time for a region with 5, 000 SNPs. To assess how the number of causal SNPs affects the efficiency of our algorithm, we generated



**Fig 4. BEAVR is computationally efficient.** (A) We show the run-time in terms of seconds per iteration of the Gibbs sampler (log-scale). We compare the version of BEAVR with the algorithmic speedup outlined in <u>Materials and methods</u> ('speedup') versus the straightforward implementation ('baseline'). We vary the number of SNPs in the region while fixing the polygenicity of each region to  $p_r = 0.01$ . (B) We show the runtime of the sampler when the number of SNPs in the region is fixed to M = 1,000 and we vary the polygenicity.

https://doi.org/10.1371/journal.pcbi.1009483.g004

Trait	%regions	h <sup>2</sup> <sub>GW</sub>	р	M <sub>c</sub>	p-GENESIS
BMI	66.2	0.30 (0.004)	(0.017, 0.018)	$(7.67 \times 10^3, 8.41 \times 10^3)$	0.014(0.003)
Height	79.6	0.64 (0.004)	(0.030, 0.032)	$(1.37 \times 10^4, 1.45 \times 10^4)$	0.009(0.002)
Waist-hip ratio	40.6	0.18 (0.004)	(0.007, 0.008)	$(3.12 \times 10^3, 3.57 \times 10^3)$	0.009(0.003)
Diastolic blood pressure	35.5	0.16 (0.004)	(0.006, 0.007)	$(2.54 \times 10^3, 3.01 \times 10^3)$	-
Systolic blood pressure	34.9	0.17 (0.004)	(0.006, 0.007)	$(2.58 \times 10^3, 3.01 \times 10^3)$	-

#### Table 1. Genome-wide estimates of polygenicity and total SNP heritability.

We report the percentage of 6-Mb regions containing at least one causal SNP under the column '% regions'. Genome-wide estimates of polygenicity and heritability were computed by aggregating estimates across all regions. The standard error is reported for genome-wide heritability estimates. Here, p denotes the proportion of causal SNPs and  $M_c$  denotes the total number of causal SNPs (we report the 95% posterior credible interval for each of these parameters). The last column denotes the estimates of polygenicity computed in a previous study [2] (twice the standard error is reported in parentheses). We omit estimates for traits not available in the prior study.

https://doi.org/10.1371/journal.pcbi.1009483.t001

simulated GWAS data for 1,000 SNPs and varied the regional polygenicity from  $p_r = 0.005$ , 0.01, 0.02, 0.03, 0.04, 0.05 and observe efficiency gains across the range of parameters (Fig 4B). The optimization of our method makes it possible to efficiently analyze regions of various sizes as well as densely imputed regions with thousands of variants.

#### Contrasting genome-wide and regional polygenicity across complex traits

We applied BEAVR to estimate regional polygenicity from marginal effect size estimates for five anthropometric and blood pressure traits from the UK Biobank (see Table 1). Marginal association statistics were computed for each of these traits from a subset of unrelated individuals identified as White British (see Materials and methods). We applied BEAVR by dividing the genome into a total of 470 6-Mb regions where each region has on average 1, 000 SNPs. Since BEAVR requires an estimate of LD between the SNPs, we used in-sample LD, *i.e.*, LD computed on the White British subset of the UK Biobank. We additionally used HESS [12] to estimate regional heritability. Since BEAVR produces a posterior distribution of the regional polygenicity, we report a region to have nonzero polygenicity if the posterior mean—(2× posterior standard deviations) does not overlap 0. Furthermore, we estimate the genome-wide polygenicity for a trait as the sum of the posterior means of regional polygenicity across all regions.

Consistent with previous estimates of genome-wide polygenicity [2], we observe that all the analyzed traits are highly polygenic. Across the traits, we observe that over one-third of the regions in the genome contain at least one causal SNP and overall each of the traits is estimated to harbor at least 1, 000 causal SNPs (Table 1). We also observe variation across traits: for height, nearly 80% of the regions contain at least one causal SNP and the total number of causal SNPs could be as high as 15, 000 while blood pressure traits are estimated to harbor about 2, 500 - 3, 000 causal SNPs. Our estimates for the proportion of causal SNPs for height is significantly higher than previously reported [2] (Table 1): the 95% credible interval estimated by BEAVR is (3.0%, 3.2%) while the estimates from prior work [2] are 0.9% with standard error 0.1%. We hypothesize that this difference is due, in part, to our method capturing smaller effect sizes by fully modeling LD, which is consistent with our simulations, but could also arise from the differences in SNP sets and GWAS summary statistics analyzed.

Previous studies have used the proportion of genomic regions with nonzero heritability as a proxy for polygenicity since nonzero heritability requires at least one causal SNP in the region [12, 21]. However, the distribution of regional heritability does not fully reflect the distribution of regional polygenicity (Fig 5, S4 and S5 Figs). Across the traits, the proportion of regions



Fig 5. Distribution of regional polygenicity and heritability. We divide the genome into 6-Mb regions and report the posterior mean of the regional polygenicity for each region across height and diastolic blood pressure. Using external software [12], we report the distribution of regional heritability for each trait.

containing at least one causal SNP is substantially higher than the estimated proportion of causal SNPs across the genome (Table 1): while  $\approx$  80% of regions contain at least one causal SNP for height, we estimate that  $\approx$  3% of the SNPs are casual. Further, we observe wide variation in regional polygenicity where, across the analyzed traits, nearly 50% of regions contain at least 5 causal SNPs and about 5% of regions contain at least 50 causal SNPs (Fig 5 and S5 Fig). These results demonstrate the additional information that can be obtained from estimates of regional polygenicity.

#### Heritability is proportional to the number of causal SNPs

Previous studies have documented a linear relationship between chromosome length and the per-chromosome heritability for multiple traits suggesting that the architecture of these traits is highly polygenic [12, 31]. We replicate this relationship between the number of SNPs and the heritability in a genomic region for each trait (*p*-value =  $1.22 \times 10^{-13}$ ;  $R^2 = 0.162$  averaged across traits; Table 2). In addition, we observe that a linear regression of heritability on the number of causal SNPs in the region is significant (*p*-value =  $2.60 \times 10^{-21}$ ;  $R^2 = 0.278$  averaged across traits) (Table 2). We also observe that the number of causal SNPs in a region better explains regional heritability than the number of overall SNPs in the region. This ranges from approximately the same  $R^2$  in systolic blood pressure to nearly three times in WHR (Table 2).

Trait	$R^2(h_r^2 \sim M_r)$	$R^2(h_r^2 \sim M_{Cr})$	$R^2(h_r^2 \sim M_r + M_{Cr})$	$R^2(h_r^2 \sim all\text{-annotations})$
BMI	$0.182 (3.67 \times 10^{-22})$	$0.226 (7.64 \times 10^{-28})$	$0.347~(6.38 \times 10^{-44})$	$0.532 (1.20 \times 10^{-27})$
Height	$0.172 (5.34 \times 10^{-21})$	$0.447 (2.96 \times 10^{-62})$	$0.501 (3.30 \times 10^{-71})$	$0.647 (3.23 \times 10^{-47})$
Waist-hip ratio	$0.105~(6.13 \times 10^{-13})$	$0.295 (2.05 \times 10^{-37})$	$0.352 (1.08 \times 10^{-44})$	$0.540 (8.23 \times 10^{-29})$
Diastolic blood pressure	$0.183 (2.60 \times 10^{-22})$	$0.254 (1.08 \times 10^{-31})$	$0.359(7.33 \times 10^{-46})$	$0.530 (2.09 \times 10^{-27})$
Systolic blood pressure	$0.168 (2.09 \times 10^{-20})$	$0.169 (1.30 \times 10^{-20})$	$0.311 (1.77 \times 10^{-38})$	$0.532 (1.11 \times 10^{-27})$

Table 2. Linear relationship between heritability, number of SNPs, number of causal SNPs, and genomic annotations.

In the first column, we model the linear relationship between the heritability of a trait and the number of SNPs across all regions of the genome. We report the coefficient of determination ( $R^2$ ). The relationship is significant for all traits (*p*-values are reported in parentheses). We observe a similar trend relating the heritability and number of causal SNPs in a region. We perform a multivariate regression to assess the relationship between the heritability and both the number of SNPs and number of causal SNPs in a region. Finally, in the last column, we perform a multivariate regression of heritability on the number of SNPs, number of causal SNPs, number of genes, median *B* value, and functional annotations [28]. Significant annotations are listed in Table B in S1 File.

https://doi.org/10.1371/journal.pcbi.1009483.t002

The slope of the regression of regional heritability on the number of causal SNPs averaged across traits is  $1.63 \times 10^{-5}$ , which can be interpreted as the heritability per additional causal SNP (Table A in S1 File). Performing multiple regression, we find that both the number of SNPs and number of causal SNPs have a significant relationship to the heritability in a region (average *p*-value =  $3.60 \times 10^{-39}$ ;  $R^2 = 0.374$ ). We hypothesize that the number of SNPs and causal SNPs together explain more of the variation in heritability than the number of causal SNPs alone due, in part, to inaccurate estimates of the number of causal SNPs and regional heritability as well as possible misspecifications in the model assumed by BEAVR.

We further investigate the relationship between genomic annotations and heritability as well as the number of causal SNPs in a region. Including the number of genes, median *B* value, and functional annotations [28] as covariates in the regression (see Materials and methods), only the number of causal SNPs remains significant (average *p*-value =  $6.37 \times 10^{-11}$ , *p*-value < 0.05/(number of annotations)) while the total number of SNPs in the region remains significant for 3 out of 5 traits (Table B in <u>S1 File</u>). None of the other genomic annotations are significant after the multiple testing correction.

While the expected regional heritability can be partly explained by the number of causal SNPs, we also observe regions that have disproportionately high heritability given the number of estimated causal SNPs (Fig 6 and S6 Fig and Table A in S1 File). These outlier regions (defined as regions with an absolute studentized residual larger than 3) are likely to harbor SNPs with larger effect sizes compared to other regions. Consistent with this hypothesis, 24 out of 27 outlier regions contain at least one genome-wide significant SNP for at least one trait. This proportion is significantly higher than a randomly chosen set of 27 regions (*p*-value  $< \frac{1}{1,000}$ ). Taken together, our analyses indicate that the heritability of a trait is composed of a mixture of small effect SNPs as well as some SNPs with relatively larger effects.

Finally, we also investigate whether the gene density in a region plays a role in the observed regional polygenicity estimates. We perform a likelihood ratio test between the following two models to assess the effect of gene density on the number of causal SNPs  $(M_{C_r})$  after adjusting for both regional heritability and the number of SNPs:  $H_0: M_{C_r} \sim h_r^2 + M_r; H_1: M_{C_r} \sim h_r^2 + M_r + \#$ genes. As shown in Table C in S1 File, we find that only the likelihood ratio test for height is significant after adjusting for the number of tested traits (*p*-value  $< \frac{0.05}{5}$ ). This observation could be due to the fact that we included all protein coding genes in the analysis regardless of the specific biological mechanism of each gene. For example, when analyzing BMI, one would expect regions with genes related to



**Fig 6. Heritability is proportional to the number of causal SNPs in a region.** We show the relationship between the number of causal SNPs and heritability for each region across height and diastolic blood pressure. We fit a linear regression for each trait and report the slope of the regression, which can be interpreted as the increase of heritability per additional causal SNP. Horizontal error bars represent two posterior standard deviations around our estimates for the number of causal SNPs. Vertical error bars represent twice the standard error around the estimates of regional heritability. Dots in black denote outlier regions which have an absolute studentized residual larger than 3.

lipids or metabolism to harbor more causal variants than genes related to seemingly unrelated biological mechanisms. This observed effect of gene density on the polygenicity of height is consistent with the hypothesis that genes throughout the genome, regardless of the specific biological function, contribute to the variance of height. Previous work has shown that height is one of the most polygenic traits with numerous causal variants spread throughout the genome [15]. This idea that numerous genes, regardless of functional mechanism, have an effect on a trait is related to the recently proposed 'omnigenic' model [5].

### Discussion

In this work, we propose BEAVR, a novel, scalable method to estimate regional polygenicity from GWAS effect size estimates in a Bayesian framework. We employ a fast inference algorithm that enables efficient inference while fully accounting for LD. Applying BEAVR to anthropometric and blood pressure traits in the UK Biobank, we observe that all of the analyzed traits are highly polygenic. At least a third of 6-Mb regions harbor at least one causal variant with this fraction rising as high as 80% for height. We find that the proportion of regions containing at least one causal SNP, which is often used as a proxy for polygenicity in previous studies, is much higher than our estimates of the proportion of causal SNPs. Additionally, we observe wide variation in regional polygenicity with an average of 48.9% of regions across the analyzed traits containing at least 5 causal SNPs and 5.44% of regions containing at least 50 causal SNPs. Finally, we find that the number of causal SNPs better explains variation in SNP heritability across regions compared to the total number of SNPs.

The observed polygenic architecture of complex traits supports the hypothesis that the majority of trait variation is modulated by variants distributed across the genome. Trait heritability is largely driven by the number of causal variants and most of these variants are spread uniformly across the genome. This finding suggests that a large proportion of genes have at least some, although limited, effect on a trait. These findings are consistent with the recently

proposed omnigenic model which suggests that disease risk is driven by a combination of a small number of primary 'core' genes and numerous 'peripheral' genes which are connected to core genes via highly interconnected gene networks [5].

We conclude by discussing limitations of our study and directions for future work. First, our model assumes that the causal effects are drawn from a single Gaussian distribution. This assumption can be relaxed and other distributions (such as mixtures of Gaussians) can be used instead. Second, our estimates of genome-wide polygenicity assume that the LD matrix is block structured which allows us to estimate genome-wide polygenicity by applying our method to regions corresponding to LD blocks. Finally, our analyses in the UK Biobank were limited to array data and thus the set of SNPs used in our analyses are missing true causal SNPs that were not typed. We leave a more thorough investigation of this scenario and analyses on imputed data as future work.

# Supporting information

**S1 Fig. A graphical model describing the generative process of our data.** Directed graphical model diagram for BEAVR. (TIF)

S2 Fig. Inferring regional polygenicity from non-Gaussian causal effect size distributions. We simulate effect sizes from mixture of Gaussian distributions with the following set of variance components:  $[1 \times 10^{-3}, 1 \times 10^{-4}]$ ;  $[1 \times 10^{-4}, 1 \times 10^{-5}]$ ;  $[1 \times 10^{-3}, 1 \times 10^{-4}, 1 \times 10^{-5}]$ . The polygenicity of the region equals the sum of the mixture proportions and the number of causal SNPs are spread equally amongst all the mixture components. For causal effect sizes drawn from the distribution with larger variances (e.g.  $1 \times 10^{-3}$ ), our estimates are relatively unbiased. However, for distributions with smaller variance components (e.g.  $1 \times 10^{-5}$ ), we start to see a downward bias proportional to the fraction of SNPs drawn from the distribution with the smaller variance component(s). (TIF)

S3 Fig. Assessing the role of reference LD in estimating regional polygenicity. The first GWAS (left) is simulated LD computed with genotypes from the UK Biobank (N = 337, 205) and inference is performed using LD computed from the European individuals from the 1000 Genomes reference panel (N = 503). The second GWAS (right) is simulated with LD derived from a subset of ( $N_1 = 168, 602$ ) genotypes from the UK Biobank and inference is performed using LD computed from a separate, non-overlapping subset ( $N_2 = 168, 602$ ) of individuals also from the UK Biobank. We find that when using LD from separate studies (1000 Genomes), BEAVR fails to accurately estimate the regional polygenicity. However, when we use LD computed from a separate set of individuals from the same study, we find our estimates are approximately unbiased.

(TIF)

**S4 Fig. Distribution of regional heritability.** We divide the genome into 6-Mb regions and report the posterior mean of the regional polygenicity for each region across BMI, waist-hip ratio (WHR), and systolic blood pressure. (TIF)

**S5 Fig. Distribution of regional polygenicity.** Using external software [12], we estimate the heritability in each 6-Mb region for each trait. We report the distribution of regional heritability for BMI, waist-hip ratio (WHR), and systolic blood pressure. (TIF)

**S6 Fig. Heritability is proportional to the number of causal SNPs in a region.** We show the relationship between the number of causal SNPs and heritability for each region across BMI, waist-hip ratio (WHR), and systolic blood pressure. We fit a linear regression for each trait. Horizontal error bars represent two posterior standard deviations around our estimates for the number of causal SNPs. Vertical error bars represent twice the standard error around the estimates of regional heritability. Dots in black denote outlier regions which have an absolute studentized residual larger than 3.

(TIF)

S1 File. Supplementary materials. Additional derivations for the Gibbs sampler. Table A. Linear relationship between the number of causal SNPs and heritability. We model the linear relationship between the number of causal SNPs for a trait and the heritability across all regions of the genome. We report the slope of the regression and the standard error. The slope can be interpreted as the expected per-SNP heritability contribution per causal SNP. The last column reports the number of 'outlier' regions for each trait, defined as a region with an absolute studentized residual greater than 3. Table B. Covariates that are associated with regional heritability  $h_{2}^{2}$ . We perform a multivariate regression of heritability on the number of SNPs, number of causal SNPs, number of genes, median B-statistic, and non-cell-type-specific annotations [28]. Only the number of causal SNPs ( $M_c$ ) remains significant for all traits after the multiple testing correction (average *p*-value =  $6.37 \times 10^{-11}$ ), and the number of SNPs ( $M_r$ ) remains significant for 3 our of 5 traits after the multiple testing correction. Table C. Likelihood ratio test assessing the role of gene density in regional polygenicity estimates. We perform a likelihood ratio test between the following two models to assess the effect of gene density on the number of causal SNPs  $(M_c)$  after adjusting for both regional heritability and the number of SNPs  $(H_0:M_{C_r}\sim h_r^2+M_r;H_1:M_{C_r}\sim h_r^2+M_r+\texttt{#genes}).$ (PDF)

# **Acknowledgments**

We are grateful to Nicholas Mancuso, Rob Brown, Alec Chiu for their helpful and insightful discussions.

# **Author Contributions**

Conceptualization: Bogdan Pasaniuc, Sriram Sankararaman.

Data curation: Kathryn S. Burch, Kangcheng Hou.

Formal analysis: Ruth Johnson, Kangcheng Hou, Mario Paciuc.

Funding acquisition: Bogdan Pasaniuc, Sriram Sankararaman.

Investigation: Ruth Johnson, Mario Paciuc.

Methodology: Ruth Johnson, Sriram Sankararaman.

Software: Ruth Johnson.

Supervision: Bogdan Pasaniuc, Sriram Sankararaman.

Validation: Ruth Johnson, Mario Paciuc.

Writing – original draft: Ruth Johnson, Kathryn S. Burch, Bogdan Pasaniuc, Sriram Sankararaman.

Writing – review & editing: Ruth Johnson, Kathryn S. Burch, Bogdan Pasaniuc, Sriram Sankararaman.

### References

- Chatterjee N, Wheeler B, Sampson J, Hartge P, Chanock SJ, Park JH. Projecting the performance of risk prediction based on polygenic analyses of genome-wide association studies. Nature genetics. 2013; 45(4):400. https://doi.org/10.1038/ng.2579 PMID: 23455638
- Zhang Y, Qi G, Park JH, Chatterjee N. Estimation of complex effect-size distributions using summarylevel statistics from genome-wide association studies across 32 complex traits. Nature genetics. 2018; 50(9):1318. https://doi.org/10.1038/s41588-018-0193-x PMID: 30104760
- Zeng J, Vlaming R, Wu Y, Robinson MR, Lloyd-Jones LR, Yengo L, et al. Signatures of negative selection in the genetic architecture of human complex traits. Nature genetics. 2018; 50(5):746. https://doi. org/10.1038/s41588-018-0101-4 PMID: 29662166
- O'Connor LJ, Schoech AP, Hormozdiari F, Gazal S, Patterson N, Price AL. Extreme polygenicity of complex traits is explained by negative selection. The American Journal of Human Genetics. 2019; 105(3):456–476. https://doi.org/10.1016/j.ajhg.2019.07.003 PMID: 31402091
- Boyle EA, Li YI, Pritchard JK. An expanded view of complex traits: from polygenic to omnigenic. Cell. 2017; 169(7):1177–1186. https://doi.org/10.1016/j.cell.2017.05.038 PMID: 28622505
- Liu X, Li YI, Pritchard JK. Trans effects on gene expression can drive omnigenic inheritance. Cell. 2019; 177(4):1022–1034. https://doi.org/10.1016/j.cell.2019.04.014 PMID: 31051098
- Moser G, Lee SH, Hayes BJ, Goddard ME, Wray NR, Visscher PM. Simultaneous discovery, estimation and prediction analysis of complex traits using a Bayesian mixture model. PLoS genetics. 2015; 11(4): e1004969. https://doi.org/10.1371/journal.pgen.1004969 PMID: 25849665
- Lloyd-Jones LR, Zeng J, Sidorenko J, Yengo L, Moser G, Kemper KE, et al. Improved polygenic prediction by Bayesian multiple regression on summary statistics. Nature communications. 2019; 10(1):1–11. https://doi.org/10.1038/s41467-019-12653-0 PMID: 31704910
- Zhou X, Carbonetto P, Stephens M. Polygenic modeling with Bayesian sparse linear mixed models. PLoS genetics. 2013; 9(2):e1003264. https://doi.org/10.1371/journal.pgen.1003264 PMID: 23408905
- Hu Y, Lu Q, Powles R, Yao X, Yang C, Fang F, et al. Leveraging functional annotations in genetic risk prediction for human complex diseases. PLoS computational biology. 2017; 13(6):e1005589. <u>https:// doi.org/10.1371/journal.pcbi.1005589</u> PMID: 28594818
- Hu Y, Lu Q, Liu W, Zhang Y, Li M, Zhao H. Joint modeling of genetically correlated diseases and functional annotations increases accuracy of polygenic risk prediction. PLoS genetics. 2017; 13(6): e1006836. https://doi.org/10.1371/journal.pgen.1006836 PMID: 28598966
- Shi H, Kichaev G, Pasaniuc B. Contrasting the genetic architecture of 30 complex traits from summary association data. The American Journal of Human Genetics. 2016; 99(1):139–153. <u>https://doi.org/10. 1016/j.ajhg.2016.05.013</u> PMID: 27346688
- Hou K, Burch KS, Majumdar A, Shi H, Mancuso N, Wu Y, et al. Accurate estimation of SNP-heritability from biobank-scale data irrespective of genetic architecture. Nature genetics. 2019; 51(8):1244–1251. https://doi.org/10.1038/s41588-019-0465-0 PMID: 31358995
- Bulik-Sullivan BK, Loh PR, Finucane HK, Ripke S, Yang J, Patterson N, et al. LD Score regression distinguishes confounding from polygenicity in genome-wide association studies. Nature genetics. 2015; 47(3):291–295. https://doi.org/10.1038/ng.3211 PMID: 25642630
- Yang J, Benyamin B, McEvoy BP, Gordon S, Henders AK, Nyholt DR, et al. Common SNPs explain a large proportion of the heritability for human height. Nature genetics. 2010; 42(7):565. https://doi.org/ 10.1038/ng.608 PMID: 20562875
- Guan Y, Stephens M, et al. Bayesian variable selection regression for genome-wide association studies and other large-scale problems. The Annals of Applied Statistics. 2011; 5(3):1780–1815. <u>https://doi.org/ 10.1214/11-AOAS455</u>
- Mitchell TJ, Beauchamp JJ. Bayesian Variable Selection in Linear Regression. Journal of the American Statistical Association. 1988; 83(404):1023–1032. https://doi.org/10.1080/01621459.1988.10478694
- Hormozdiari F, Kostem E, Kang EY, Pasaniuc B, Eskin E. Identifying causal variants at loci with multiple signals of association. Genetics. 2014; p. genetics–114. PMID: 25104515
- Zhu X, Stephens M. Bayesian large-scale multiple regression with summary statistics from genomewide association studies. The annals of applied statistics. 2017; 11(3):1561. <u>https://doi.org/10.1214/17-AOAS1046 PMID: 29399241</u>
- 20. Brooks S, Gelman A, Jones G, Meng XL. Handbook of markov chain monte carlo. CRC press; 2011.

- Loh PR, Bhatia G, Gusev A, Finucane HK, Bulik-Sullivan BK, Pollack SJ, et al. Contrasting genetic architectures of schizophrenia and other complex diseases using fast variance-components analysis. Nature genetics. 2015; 47(12):1385. https://doi.org/10.1038/ng.3431 PMID: 26523775
- 22. Geman S, Geman D. Stochastic relaxation, Gibbs distributions, and the Bayesian restoration of images. IEEE Transactions on pattern analysis and machine intelligence. 1984;(6):721–741. <u>https://doi.org/10.1109/TPAMI.1984.4767596 PMID: 22499653</u>
- Bycroft C, Freeman C, Petkova D, Band G, Elliott LT, Sharp K, et al. The UK Biobank resource with deep phenotyping and genomic data. Nature. 2018; 562(7726):203. <u>https://doi.org/10.1038/s41586-018-0579-z PMID: 30305743</u>
- Lee JJ, Wedow R, Okbay A, Kong E, Maghzian O, Zacher M, et al. Gene discovery and polygenic prediction from a 1.1-million-person GWAS of educational attainment. Nature genetics. 2018; 50(8):1112.
- Linnér RK, Biroli P, Kong E, Meddens SFW, Wedow R, Fontana MA, et al. Genome-wide association analyses of risk tolerance and risky behaviors in over 1 million individuals identify hundreds of loci and shared genetic influences. Nature genetics. 2019; 51(2):245. <u>https://doi.org/10.1038/s41588-018-0309-3</u>
- Consortium GP, et al. A global reference for human genetic variation. Nature. 2015; 526(7571):68. https://doi.org/10.1038/nature15393
- Chang CC, Chow CC, Tellier LC, Vattikuti S, Purcell SM, Lee JJ. Second-generation PLINK: rising to the challenge of larger and richer datasets. Gigascience. 2015; 4(1):s13742–015. PMID: 25722852
- Finucane HK, Bulik-Sullivan B, Gusev A, Trynka G, Reshef Y, Loh PR, et al. Partitioning heritability by functional annotation using genome-wide association summary statistics. Nature genetics. 2015; 47(11):1228. https://doi.org/10.1038/ng.3404 PMID: 26414678
- Freund MK, Burch KS, Shi H, Mancuso N, Kichaev G, Garske KM, et al. Phenotype-specific enrichment of Mendelian disorder genes near GWAS regions across 62 complex traits. The American Journal of Human Genetics. 2018; 103(4):535–552. https://doi.org/10.1016/j.ajhg.2018.08.017 PMID: 30290150
- McVicker G, Gordon D, Davis C, Green P. Widespread genomic signatures of natural selection in hominid evolution. PLoS genetics. 2009; 5(5). <u>https://doi.org/10.1371/journal.pgen.1000471</u> PMID: 19424416
- Yang J, Manolio TA, Pasquale LR, Boerwinkle E, Caporaso N, Cunningham JM, et al. Genome partitioning of genetic variation for complex traits using common SNPs. Nature genetics. 2011; 43(6):519. https://doi.org/10.1038/ng.823 PMID: 21552263